Title: Emerging serotype III Sequence type 17 group B streptococcus invasive infection in infants: the clinical characteristics and impacts on outcomes

Authors:

Yi Kao (cathaykao@gmail.com)
Ming-Horng Tsai (mingmin.tw@yahoo.com.tw)
Mei-Yin Lai (lmi818@msn.com)
Shih-Ming Chu (kz6479@cgmh.org.tw)
Hsuan-Rong Huang (qbonbon@gmail.com)
Ming-Chou Chiang (cmc123@cgmh.org.tw)
Ren-Huei Fu (rkenny@cgmh.org.tw)
Jang-Jih Lu (janglu45@gmail.com)
Jen-Fu Hsu (jeff0724@gmail.com)

Version: 2 Date: 01 Jun 2019

Author’s response to reviews:

Editor Comments:

1. We noticed that the corresponding author information provided on your title page is different to the one you indicated in the editorial submission. Please amend one so that they match.

Reply: Thank you for your instructive advice. I will make them match, thank you.

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Reply: Thank you for your instructive advice. I will upload my clean version of final manuscript, thank you.